\begin{table}%t1
\caption{\label{tab_b2}The sample of low redshift B2 radio galaxies with velocity dispersion measurements.}     %\centerline
{\begin{tabular}{lcccccccc }        
\hline\hline
Name & $m_V$ &    $ z$   &    $\sigma_{\rm c}$ & $\Delta$$\sigma$ &  $\langle\mu_{\rm b}\rangle$ &  $\Delta$$\langle\mu_{\rm b}\rangle$ & $R_{\rm b}$  & $\Delta$$r_{\rm b}$ \\
(1) & (2) & (3) & (4) & (5) & (6) & (7) & (8) & (9) \\
\hline        
0648+27$^+$   &  13.82  &  0.0409 &  137.8  & 38.0 & 14.94 & 0.05 & 0.26 & 0.02  \\                
0755+37    &  13.80  &  0.0413 &  291.0  & 17.3 & 17.05 & 0.15 & 1.09 & 0.09  \\
0908+37    &  15.67  &  0.1040 &  387.1  & 18.0 & 17.36 & 0.10 & 0.50 & 0.05  \\
0915+32B &  15.20  & 0.0620 &  252.1  & 24.3 & 17.00 & 0.05 & 0.52 & 0.03 \\
0924+30    &  13.32  & 0.0266 &  243.7  & 14.5 & 17.33 & 0.15 & 0.96 & 0.14 \\
1003+26    &  15.47  & 0.1165 &  438.4  & 21.6 & 17.94 & 0.15 & 0.41 & 0.05 \\
1113+24    &  15.10  & 0.1021 &  377.8  & 10.2 & 17.44 & 0.05 & 0.47 & 0.03 \\
1204+34    &  15.35  & 0.0788 &  201.0  & 29.6 & 17.47 & 0.25 & 0.46 & 0.09 \\
1322+36B &  12.84  & 0.0175 &  259.1  & 12.1 & 15.45 & 0.05 & 0.62 & 0.05 \\
1339+26B  &  15.10 & 0.0757 &  379.1  & 18.2 & 16.41 & 0.15 & 0.32 & 0.04 \\
1347+28    &  15.66  & 0.0724 &  228.0  & 19.4 & 17.18 & 0.05 & 0.34 & 0.03 \\
1357+28    &  14.81  & 0.0629 &  308.3  & 11.2 & 16.80 & 0.05 & 0.40 & 0.03  \\
1422+26B &  14.32  & 0.0370 &  264.9  & 11.6 & 15.44 & 0.05 & 0.13 & 0.01 \\
1430+25    &  16.65  & 0.0813 &  203.6  & 14.3 & 16.72 & 0.25 & 0.13 & 0.03 \\
1447+27    &  13.88  & 0.0306 &  318.4  & 11.6 & 15.77 & 0.15 & 0.47 & 0.04 \\
1450+28  &  16.31  &  0.1203  & 405.6  & 23.6 & 16.18 & 0.05 & 0.16 & 0.15 \\
1450+28*$^+$  &  16.31   & 0.1265 &  333.3  & 22.1 & 16.56 & 0.05 & 0.20 & 0.01 \\
1502+26   &  15.42   & 0.0540 &  402.1  & 19.9 & 16.60 & 0.15 & 0.66 & 0.07 \\
%1502+26B & 15.24 & 0.0540 &  390.0  & 22.7 & 16.60 & 0.15 & 0.66 & 0.07 \\
1512+30   &  15.37   & 0.0931 &  330.2  & 25.6 & 16.55 & 0.05 & 0.34 & 0.03 \\
1521+28   &  15.09   & 0.0825 &  323.0  & 11.8 & 18.02 & 0.15 & 0.85 & 0.06 \\
1527+30   &  15.64   & 0.1143 &  448.0  & 23.0 & 16.76 & 0.10 & 0.34 & 0.03 \\
1553+24   &  14.41   & 0.0426 &  269.9  & 10.9 & 16.41 & 0.05 & 0.45 & 0.03 \\
1557+26   &  14.97   & 0.0442 &  310.7  & 16.1 & 15.45 & 0.05 & 0.21 & 0.01 \\
1613+27   &  14.90   & 0.0647 &  246.6  & 13.5 & 16.74 & 0.15 & 0.30 & 0.03 \\
1658+30   &  14.47   & 0.0351 &  315.8  & 10.2 & 15.02 & 0.15 & 0.19 & 0.02 \\
1726+31   &  17.04   & 0.1670 &  294.8  & 26.8 & 15.96 & 0.15 & 0.22 & 0.03 \\
1827+32   &  15.10   & 0.0659  & 308.5  & 11.4 & 16.24 & 0.05 & 0.21 & 0.01 \\
2236+35   &  12.57   & 0.0276 & 297.8   & 12.5 & 17.09 & 0.05 & 1.28 & 0.11  \\
\hline    
0034+25 &  13.35 &  0.0318   &   295.0   & 10.0 & 16.27 & 0.05 & 0.74 & 0.04 \\
0055+26 &  13.80 &  0.0474   &   231.0   & 13.0 & 16.34 & 0.05 & 0.40 & 0.02 \\
0120+33  & 11.43  &  0.0165   &   315.0   & 10.0 & 15.13 & 0.05 & 0.75 & 0.07 \\
1217+29 &  10.41 &  0.0022   &   238.0   & 10.0 & 13.87 & 0.05 & 0.80 & 0.06 \\
1256+28 &  13.88 &  0.0229   &   203.0   & 10.0 & 16.48 & 0.15 & 0.70 & 0.07 \\
1257+28 &  11.94 &  0.0241   &   278.0   & 10.0 & 17.97 & 0.10 & 2.55 & 0.01 \\
1525+29 &  14.95  &  0.0652   &   250.0   & 30.0 & 16.56 & 0.15 & 0.53 & 0.05 \\
1610+29 &  13.29 &  0.0318   &   331.0   & 26.0 & 16.49 & 0.05 & 0.95 & 0.06 \\
3C88       &  14.24  & 0.0302    &   190.0   & 22.0 & 17.47 & 0.05 & 0.98 & 0.09 \\
3C192     &  15.72  & 0.0597    &   199.0   & 23.0 & 14.34 & 0.15 & 0.09 & 0.01 \\
3C388     &  14.77 & 0.0917    &   365.0   & 23.0 & 18.03 & 0.15 & 0.96 & 0.09 \\
\hline                                   
\end{tabular}}
\medskip
\par
Columns: (1) Identification of the source, (2) apparent magnitude $m_V$ from \cite{col75} (photographic magnitudes converted to visual magnitudes,
\cite{fan78}, \cite{gon93}, and \cite{gon00}, (3) redshift, 
(4, 5) measured velocity dispersion and error in km~s$^{-1}$, (8,~9), $\langle\mu_{\rm b}\rangle$ and error in mag/arcsec$^2$, (8, 9) break radius $R_{\rm b}$ and error in arcsecs,  from \cite{hans1};~
 *companion of the radio galaxy, + not used in the CFP see Appendix~A.1.\vspace*{2.4mm}
\end{table}